Author's response to reviews

Title: CT60 genotype does not affect CTLA-4 isoform expression despite association to autoimmune disease in northern Sweden

Authors:

Sofia Mayans (sofia.mayans@medbio.umu.se)
Kurt Lackovic (kurt.lackovic@medbio.umu.se)
Caroline Nyholm (caroline.nyholm@med.lu.se)
Petter Lindgren (petter.lindgren@medbio.umu.se)
Karin Ruikka (karin.ruikka@nll.se)
Mats Eliasson (mats.eliasson@nll.se)
Corrado M Cilio (corrado.cilio@med.lu.se)
Dan Holmberg (dan.holmberg@medbio.umu.se)

Version: 2 Date: 21 August 2006

Author's response to reviews:

Dear Sir/Madam,

Thank you for the comments on our manuscript. We agree with the suggestion that it would be better suited to BMC Medical Genetics and have therefore revised it according to your suggestions. The alterations include:

Restructuring of the Abstract and Manuscript sections to concur with the format of BMC Med Gen.

We have also included a "Competing Interest" section in the manuscript.

With these alterations we hope that the manuscript can now be considered for publication in BMC Medical Genetics.

Sincerely yours

Dan Holmberg